Author's response to reviews

Title: Population Attributable Fraction of Smoking and Metabolic Syndrome on Cardiovascular Disease Mortality in Japan: a 15-Year Follow Up of NIPPON DATA90.

Authors:

Naoyuki Takashima (takasima@belle.shiga-med.ac.jp)
Katsuyuki Miura (miura@belle.shiga-med.ac.jp)
Atsushi Hozawa (ahozawa@med.id.yamagata-u.ac.jp)
Aya Kadota (ayakd@belle.shiga-med.ac.jp)
Tomonori Okamura (okamurat@hsp.ncvc.go.jp)
Yasuyuki Nakamura (nakamury@kyoto-wu.ac.jp)
Takehito Hayakawa (thayaka@fmu.ac.jp)
Nagako Okuda (okuda@belle.shiga-med.ac.jp)
Akira Fujiyoshi (afujiy@belle.shiga-med.ac.jp)
Shin-ya Nagasawa (nagasawa@belle.shiga-med.ac.jp)
Takashi Kadowaki (kadowaki@belle.shiga-med.ac.jp)
Yoshitaka Murakami (ymura@belle.shiga-med.ac.jp)
Yoshikuni Kita (kita@belle.shiga-med.ac.jp)
Akira Okayama (aokayama@jatahq.org)
Hirotsugu Ueshima (hueshima@belle.shiga-med.ac.jp)
NIPPON DATA90 Research Group [] (hueshima@belle.shiga-med.ac.jp)

Version: 4 Date: 24 May 2010

Author's response to reviews: see over
To the Editor:

We have revised the manuscript according to the comments. Points of the revisions are summarized as follows:

**Major revisions (we require the author to make these changes)**

**Author groups:** please include the name of the group within the list of authors, by adding the name to the ?first name? box; and the bracketed abbreviation to the ?last name? box in the submission system.

Please remove the names of the members of the group from the title page and place them in the Acknowledgements, as they are to be acknowledged as members of the study group.

The groups?’ contribution as a whole can be given in the authors’ contributions section.

We included the name of group within the list of authors.

Most of members of the group did not include the list of authors. Some of the members of the group were included authors list. They contributed to the concept, design, analysis, interpretation of data, and preparation of the manuscript. Another members of the group contributed to the data collection and we placed them in the Acknowledgements.

**References:** Please make sure references are cited in numerical order. We notice reference 20 is not cited within the main text. Please make sure references continue to be cited in numerical order once this reference is cited.

We checked the manuscripts and deleted the reference 20.

**Tables:** tables should fit into an A4 page (maximum size 176mm x 230mm), and the orientation must be portrait only. Tables of up to 3 pages are allowed.
If you cannot re-orientate the table to fit this size, please move it to the additional files. Tables in the additional files will be made available with the final manuscript upon publication but will not be included in the manuscript in situ. Please ensure that you update the text to cite additional files and include a legend in the additional files section.

We changed tables which fit into an A4 page.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Email address: please include email addresses for all the authors on the title page, using the following format:

AB: abcd@institution.ac.uk
EF: efgh@generic.co.uk
IJ: IJKL@corporation.com

We changed the format as above.

Underlining: Please note that underlining is not supported in the final full text; you may wish to change it to bold.

We deleted underlining.

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.
We checked our manuscript.